Author's response to reviews

Title: Impact of today's media on university student's body image in Pakistan; a conservative, developing country's perspective.

Authors:

Amad N Khan (amadnaseer@gmail.com)
Salema Khalid (inzmania@hotmail.com)
Hussain I Khan (hussain.i.khan@gmail.com)
Mehnaz Jabeen (mehnaj.jabeen@aku.edu)

Version: 6 Date: 2 May 2011

Author's response to reviews: see over
Dear Editor,

We have submitted a revised manuscript after getting it reviewed and edited by a scientific writer to ensure that it conforms to standard scientific writing practices. I hope our paper is now suitable for publication in your prestigious journal.

Best wishes,

Amad Khan, Salema Khalid, Hussain Khan and Mehnaz Jabeen

Dated: 02/05/2011